Author's response to reviews

Title: Inference of synergy/antagonism between anticancer drugs from the pooled analysis of clinical trials

Authors:

Wenfeng Kang (kangwe@umdnj.edu)
Robert S DiPaola (dipaolrs@umdnj.edu)
Alexei Vazquez (vazqueal@umdnj.edu)

Version: 13 Date: 15 May 2013

Author's response to reviews:

Dear Editor,

Thank you very much for handling our submission. We are very glad to hear that both reviewers consider our work worth of publication after minor changes. Please, find below a point-by-point response to the reviewers comments and suggestions.

Sincerely yours,
Alexei Vazquez
Corresponding author

Reviewer 1
Minor essential revisions
1. Have p12 and n12 in Section "Null model for combinations of two non-interacting agents" been defined.
Response: These quantities are now defined.

2. The second last sentence in the above section could be made clearer.
Response: We have edited this sentence to make it clear.

Reviewer 2
The paper 'Inference of synergy/antagonism between anticancer drugs from the pooled analysis of clinical trials' by Kang et al. has been revised according to my suggestions. At this point I don't have any other comments.

We are glad to hear that the reviewer considers this work suitable for publication on its current form.